Author's response to reviews

Title: Prevention of Type 2 Diabetes by lifestyle intervention in an Australian primary health care setting: Greater Green Triangle (GGT) Diabetes Prevention Project

Authors:

Tiina Laatikainen (tiina.laatikainen@ktl.fi)
James A Dunbar (Director@greaterhealth.org)
Anna Chapman (anna.chapman@greaterhealth.org)
Annamari Kilkinen (annamari.kilkkinen@greaterhealth.org)
Erkki Vartiainen (erkki.vartiainen@ktl.fi)
Sami Heistaro (sami.heistaro@ktl.fi)
Ben Philpot (ben.philpot@greaterhealth.org)
Pilvikki Absetz (pilvikki.absetz@ktl.fi)
Stephen Bunker (Stephen.Bunker@med.monash.edu.au)
Adrienne O'Neil (adrienne.oneil@greaterhealth.org)
Prasuna Reddy (Director.HSR@greaterhealth.org)
James D Best (jdbest@medstv.unimelb.edu.au)
Edward D Janus (edwarddj@unimelb.edu.au)

Version: 3 Date: 12 September 2007

Author's response to reviews: see over
Dear Editor,

Thank you for the confirmation of manuscript acceptance pending formatting requirements for our manuscript entitled ‘Prevention of Type 2 Diabetes by lifestyle intervention in an Australian primary health care setting: Greater Green Triangle (GGT) Diabetes Prevention Project’.

On behalf of my fellow authors I would like to thank the reviewers for their valuable comments. Please, find attached the format revised manuscript. We have updated our manuscript to conform to the formatting requirements requested. We trust that our manuscript is now acceptable for publication.

We look forward to your response.

Yours sincerely,

Dr Annamari Kilkkinen
Senior Research Fellow - Public Health
GGT UDRH

Correspondence to:
Dr Annamari Kilkkinen, Greater Green Triangle University Department of Rural Health, PO Box 423, Warrnambool, Victoria, 3280, Australia.
Telephone: 03 556133315; fax: 03 5563144; email: annamari.kilkkinen@greaterhealth.org